### **ACC/AHA Guideline**

# ACC/AHA 2002 Guideline Update for the Management of Patients With Chronic Stable Angina—Summary Article

A Report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee on the Management of Patients With Chronic Stable Angina)

The Clinical Efficacy Assessment Subcommittee of the American College of Physicians–American Society of Internal Medicine acknowledges the scientific validity of this product as a background paper and as a review that captures the levels of evidence in the management of patients with chronic stable angina as of November 17, 2002.

#### Committee Members

Raymond J. Gibbons, MD, FACC, FAHA, Chair; Jonathan Abrams, MD, FACC, FAHA; Kanu Chatterjee, MB, FACC; Jennifer Daley, MD, FACP; Prakash C. Deedwania, MD, FACC, FAHA; John S. Douglas, MD, FACC; T. Bruce Ferguson, Jr, MD; Stephan D. Fihn, MD, MPH, FACP; Theodore D. Fraker, Jr, MD, FACC; Julius M. Gardin, MD, FACC, FAHA; Robert A. O'Rourke, MD, FACC, FAHA; Richard C. Pasternak, MD, FACC, FAHA; Sankey V. Williams, MD, MACP

#### Task Force Members

Raymond J. Gibbons, MD, FACC, Chair; Joseph S. Alpert, MD, FACC, FAHA; Elliott M. Antman, MD, FACC, Vice Chair; Loren F. Hiratzka, MD, FACC, FAHA; Valentin Fuster, MD, PhD, FACC, FAHA; David P. Faxon, MD, FACC, FAHA; Gabriel Gregoratos, MD, FACC, FAHA; Alice K. Jacobs, MD, FACC, FAHA; Sidney C. Smith, Jr, MD, FACC, FAHA

<b>Table of Contents</b>	E. Obesity162
I. Introduction160	F. Metabolic Syndrome162
II. Angiotensin Converting Enzyme	G. Hormonal Replacement Therapy162
Inhibitors160	H. Oxidative Stress163
III. Treatment of Risk Factors161	IV. Alternative Therapies for Chronic Stable Angina in
A. LDL Cholesterol161	Patients Refractory to Medical Therapy Who Are Not
B. Non-HDL Cholesterol161	Candidates for Percutaneous Intervention or
C. HDL Cholesterol162	Revascularization163
D. Triglycerides162	A. Spinal Cord Stimulation163

This document was approved by the American College of Cardiology Foundation Board of Trustees in October 2002, the American Heart Association Science Advisory and Coordinating Committee in October 2002, and the Clinical Efficacy Assessment Subcommittee of the American College of Physicians—American Society of Internal Medicine in June 2002.

The ACC/AHA Task Force on Practice Guidelines makes every effort to avoid any actual or potential conflicts of interest that might arise as a result of an outside relationship or personal interest of a member of the writing panel. Specifically, all members of the writing panel are asked to provide disclosure statements of all such relationships that might be perceived as real or potential conflicts of interest. These statements are reviewed by the parent task force, reported orally to all members of the writing panel at the first meeting, and updated as changes occur. The conflict of interest information for the writing committee members is posted on the ACC and AHA World Wide Web sites with the full-length version of the update.

When citing this document, please use the following citation format: Gibbons RJ, Abrams J, Chatterjee K, Daley J, Deedwania PC, Douglas JS, Ferguson TB Jr, Fihn SD, Fraker TD Jr, Gardin JM, O'Rourke RA, Pasternak RC, Williams SV. ACC/AHA 2002 guideline update for the management of patients with chronic stable angina—summary article: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee on the Management of Patients With Chronic Stable Angina). J Am Coll Cardiol 2003;41:159–68.

Copies: This document and the full-text guideline are available on the World Wide Web sites of the ACC (www.acc.org) and the AHA (www.americanheart.org). To obtain a single copy of this summary article published in the January 7/14, 2003, issue of *Circulation* and the January 1, 2003, issue of the *Journal of the American College of Cardiology*, call 800-253-4636 (US only) or write the American College of Cardiology, Resource Center, 9111 Old Georgetown Road, Bethesda, MD 20814-1699, and ask for reprint No. 71-0244. To purchase additional reprints: up to 999 copies, call 800-611-6083 (US only) or fax 413-665-2671; 1000 or more copies, call 410-528-4426, fax 410-528-4264, or e-mail klbradle@lww.com.

(J Am Coll Cardiol 2003;41:159-68.)

©2003 by the American College of Cardiology Foundation and the American Heart Association, Inc. PII S0735-1097(02)02848-6

	B. Enhanced External Counterpulsation	.164
	C. Laser TMR	.164
V.	Asymptomatic Patients With Known or Suspected	
	Coronary Artery Disease	.164
	A. Treatment of Risk Factors	.166
	B. Revascularization	.166
Ref	erences	.167

#### I. Introduction

The American College of Cardiology (ACC)/American Heart Association (AHA) Task Force on Practice Guidelines regularly reviews existing guidelines to determine when an update or a full revision is needed. This process gives priority to areas in which major changes in text, and particularly recommendations, are merited on the basis of new understanding or evidence. Minor changes in verbiage and references are discouraged.

The ACC/AHA/American College of Physicians–American Society of Internal Medicine (ACP-ASIM) Guidelines for the Management of Patients With Chronic Stable Angina, which were published in June 1999, have now been updated. The full-text guideline incorporating the updated material is available on the Internet (www.acc.org or www.americanheart.org) in both a track-changes version showing the changes in the 1999 guideline in strike-out (deleted text) and highlighting (new text) and a "clean" version that fully incorporates all the changes.

This summary article describes the 4 most important areas of change reflected in the update in a format that we hope can be read and understood as a stand-alone document. Interested readers are referred to the full-length version on the Internet to completely understand the location of these changes within the full-length guideline, as well as their proper context. The full-length guideline includes some additional changes that are not reflected in this summary article. All new references appear in bold-faced type; all original references appear in normal type.

Although the primary focus of this guideline is on symptomatic patients, asymptomatic patients with known or suspected coronary disease are included in this update and are described in Section V.

The customary ACC/AHA classifications I, IIa, IIb, and III and the corresponding levels of evidence are used throughout the document.

- Class I: Conditions for which there is evidence or general agreement that a given procedure or treatment is useful and effective.
- Class II: Conditions for which there is conflicting evidence or a divergence of opinion about the usefulness/efficacy of a procedure or treatment.
  - Class IIa: Weight of evidence/opinion is in favor of usefulness/efficacy.
  - Class IIb: Usefulness/efficacy is less well established by evidence/opinion.
  - Class III: Conditions for which there is evidence and/or general agreement that the pro-

cedure/treatment is not useful/effective and in some cases may be harmful.

Level of Evidence A: Data derived from multiple randomized clinical trials.

Level of Evidence B: Data derived from a single randomized trial or nonrandomized studies.

Level of Evidence C: Consensus opinion of experts.

The ACC/AHA Task Force on Practice Guidelines welcomes feedback on this update process and the format of this article. Please direct your comments to the Task Force in care of Dawn Phoubandith, American College of Cardiology, or via e-mail (dphouban@acc.org).

#### **II. Angiotensin Converting Enzyme Inhibitors**

New Recommendations for Angiotensin Converting Enzyme (ACE) Inhibitors Class I

- 1. ACE inhibitor in all patients with coronary artery disease (CAD)\* who also have diabetes and/or left ventricular systolic dysfunction. (Level of Evidence: A) Class IIa
  - 1. ACE inhibitor in patients with CAD\* or other vascular disease. (Level of Evidence: B)

\*Significant CAD by angiography or previous myocardial infarction.

The results of the Heart Outcomes Prevention Evaluation (HOPE) trial now confirm that use of the ACE inhibitor ramipril (10 mg/d) reduced the incidence of cardiovascular death, myocardial infarction (MI), and stroke in patients who were at high risk for, or had, vascular disease in the absence of heart failure (1). The primary outcome in HOPE was a composite of cardiovascular death, MI, and stroke. However, the results of HOPE were so definitive that each of the components of the primary outcome by itself also showed statistical significance. Furthermore, only a small part of the benefit could be attributed to a reduction in blood pressure (-2 to -3 mm Hg).

The results of HOPE were extremely impressive when one considers the magnitude of the difference between ramipril and placebo in the primary outcomes of cardiovascular death, MI, and stroke. The HOPE study was unique in that of the 9541 patients in the study, 3577 (37.5%) had diabetes. There was a very significant reduction in diabetic complications, a composite for the development of diabetic nephropathy, need for renal dialysis, and laser therapy for diabetic retinopathy, in those patients receiving ramipril. Even more fascinating was the finding that among the patients who were not designated as diabetic at the beginning of the trial, fewer were diagnosed with diabetes during the 4-year observation period if they were treated with ramipril. Before the HOPE trial, numerous clinical trials suggested that ACE inhibitor treatment may delay or prevent cardiovascular outcomes in patients with diabetes after an MI, in the presence of hypertension, and in the presence of a low ejection fraction or heart failure. Furthermore, ACE inhibitors may also prevent

overt nephropathy and other microvascular outcomes in patients with type 1 or type 2 diabetes.

The Microalbuminuria, Cardiovascular, and Renal Outcomes (MICRO)-HOPE study, a substudy of the HOPE study, has provided new clinical data on the cardiorenal therapeutic benefits of ACE inhibitor intervention in a broad range of middle-aged patients with diabetes mellitus who are at high risk for cardiovascular events. The risk of MI was reduced by 22% (P=0.01), stroke by 33% (P=0.0074), cardiovascular death by 37% (P=0.0001), and the combined primary outcome of these events by 25% (P=0.0004). Ramipril also lowered the risk of overt nephropathy by 24% (P=0.027).

ACE inhibitors should be used in most cases as routine secondary prevention for patients with known CAD, particularly in diabetics without severe renal disease. There are 2 ongoing clinical trials evaluating the effect of 2 different ACE inhibitors in patient populations that do not include patients with diabetes mellitus. These studies will answer the question whether a vasculoprotective effect can be accomplished in a lower-risk group of patients than those enrolled in the HOPE study.

#### **III. Treatment of Risk Factors**

#### New/Changed Recommendations† Class IIa

- 1. In patients with documented or suspected CAD and low-density lipoprotein (LDL) cholesterol 100 to 129 mg/dL, several therapeutic options are available: (Level of Evidence: B)
  - a. Lifestyle and/or drug therapies to lower LDL to less than 100 mg/dL. (Level of Evidence: B)
  - b. Weight reduction and increased physical activity in persons with the metabolic syndrome. (*Level of Evidence: B*)
  - c. Institution of treatment of other lipid or nonlipid risk factors; consider use of nicotinic acid or fibric acid for elevated triglycerides or low highdensity lipoprotein (HDL) cholesterol. (Level of Evidence: B)
- Therapy to lower non-HDL cholesterol in patients with documented or suspected CAD and triglyceride levels greater than 200 mg/dL, with a target non-HDL cholesterol level of less than 130 mg/dL. (Level of Evidence: B)
- 3. Weight reduction in obese patients in the absence of hypertension, hyperlipidemia, or diabetes mellitus. (*Level of Evidence: C*)

#### Class III

- 1. Initiation of hormone replacement therapy (HRT) in postmenopausal women for the purpose of reducing cardiovascular risk. (Level of Evidence: A)
- 2. Vitamin C and E supplementation. (Level of Evidence: A)
- 3. Coenzyme Q. (Level of Evidence: C)

Many of the new recommendations reflect the latest recommendations of the National Cholesterol Education Program—Adult Treatment Panel III (ATP-III) (2), which the writing committee strongly endorses.

The results of the largest cholesterol-lowering trial yet performed, the Heart Protection Study (HPS), were published as this update was in the final stages of preparation (2a). This trial included more than 20 000 men and women age 40 to 80 years with coronary disease, other vascular disease, diabetes, and/or hypertension. Patients were randomized to simvastatin 40 mg or matching placebo and were followed for a mean of five years. The primary end point, total mortality, was reduced by statin treatment by approximately 25% overall and similarly in all important prespecified subgroups, including: women, patients more than 75 years old, diabetics, and individuals with baseline LDL cholesterol of less than 100 mg per dL. Analysis of these data by all appropriate authorities, including the National Cholesterol Education project, will be necessary to clarify their implications for these guidelines.

#### A. LDL Cholesterol

The clinical trial data indicate that in patients with established coronary disease, including chronic stable angina pectoris, dietary intervention and treatment with lipid-lowering medications should not be limited to those with extreme values. The benefits of lipid-lowering therapy were evident in patients in the lowest baseline quartile of LDL cholesterol (modest elevations) in the Scandinavian Simvastatin Survival Study (4S) and in those with minimal elevation of LDL cholesterol level in the Cholesterol And Recurrent Events (CARE) study. These trials establish the benefits of aggressive lipid-lowering treatment for most patients with coronary disease, even when LDL cholesterol is within a range considered acceptable for patients in a primary prevention setting. For patients with established coronary disease, nonpharmaceutical treatment and drug treatment are warranted in the vast majority of patients. The goal of treatment is an LDL cholesterol level less than 100 mg/dL. When LDL cholesterol is 101 to 129 mg/dL, either at baseline or with LDL-lowering therapy, several therapeutic options are available:

- Initiate or intensify lifestyle and/or drug therapies specifically to lower LDL.
- Emphasize weight reduction and increased physical activity in persons with the metabolic syndrome (see below).
- Delay use or intensification of LDL-lowering therapies and institute treatment of other lipid or nonlipid risk factors; consider use of other lipid-modifying drugs (eg, nicotinic acid or fibric acid) if the patient has elevated triglyceride or low HDL cholesterol levels.

#### B. Non-HDL Cholesterol

The finding that elevated triglycerides are an independent risk factor for coronary heart disease (CHD) suggests that some triglyceride-rich lipoproteins are atherogenic. The latter are partially degraded very-low-density lipoproteins (VLDL), commonly called "remnant lipoproteins." In clinical practice, non-HDL cholesterol is the most readily available measure of

<sup>†</sup>The original recommendations, and particularly the 6 Class I recommendations, continue to apply.

the total pool of atherogenic lipoproteins, including remnants. Thus, non-HDL cholesterol can be a target of cholesterollowering therapy. Moreover, non-HDL cholesterol is highly correlated with total apolipoprotein B (apoB) (3,4); apoB is the major apolipoprotein of all atherogenic lipoproteins. Serum total apoB also has been shown to have a strong predictive power for severity of coronary atherosclerosis and CHD events (5–12). Because of the high correlation between non-HDL cholesterol and apoB levels (3,4), non-HDL cholesterol represents an acceptable surrogate marker for total apoB; the latter is not widely available for routine measurement in clinical practice. Therefore, ATP III (2) identifies the sum of LDL and VLDL cholesterol (termed "non-HDL cholesterol" [total cholesterol minus HDL cholesterol]) as a secondary target of therapy in persons with high triglycerides (greater than 200 mg/dL). The goal for non-HDL cholesterol (for persons with serum triglycerides greater than or equal to 200) is 130 mg/dL; this is 30 mg/dL higher than the goal for LDL cholesterol, because the normal VLDL cholesterol level is 30 mg/dL.

#### C. HDL Cholesterol

ATP III has defined a low HDL cholesterol level as less than 40 mg/dL (2). Patients with established coronary disease and low HDL cholesterol are at high risk for recurrent events and should be targeted for aggressive nonpharmacological treatment (dietary modification, weight loss, and/or physical exercise). ATP III does not specify a goal for HDL raising. Although clinical trial results suggest that raising HDL will reduce risk, the evidence is insufficient to specify a goal of therapy. Furthermore, currently available drugs do not robustly raise HDL cholesterol. A low HDL level should receive clinical attention and management according to the following sequence. In all persons with low HDL cholesterol, the primary target of therapy is LDL cholesterol; ATP III guidelines for diet, exercise, and drug therapy should be followed to achieve the LDL cholesterol goal. Second, after the LDL goal has been reached, emphasis shifts to other issues. When a low HDL cholesterol level is associated with high triglycerides (200 to 499 mg/dL) secondary priority goes to achieving the non-HDL cholesterol goal, as outlined earlier. Also, if triglycerides are less than 200 mg/dL (isolated low HDL cholesterol), drugs to raise HDL (fibrates or nicotinic acid) can be considered. Nicotinic acid and fibrates usually raise HDL levels, as do 3-hydroxy-3-methylglutaryl coenzyme A (HMG-CoA) reductase inhibitors.

#### D. Triglycerides

Triglyceride levels are predictive of CHD risk in a variety of observational studies and clinical settings (13). Much of the association of triglycerides with CHD risk is related to other factors, including diabetes, obesity, hypertension, high LDL cholesterol, and low HDL cholesterol (14). In addition, hypertriglyceridemia is often found in association with abnormalities in hemostatic factors (15). Recently, however, a borderline (150 to 199 mg/dL) or high (greater than 200 mg/dL) triglyceride level has been established by meta-analyses of prospective studies as an independent risk factor for CHD (2,16,17).

Nonpharmacological management of high triglycerides consists of weight loss, reduction in alcohol consumption for those in whom this mechanism may be causal, smoking cessation, and physical activity. Drugs that can lower triglycerides include nicotinic acid, fibrate derivatives, and, to a lesser degree, statins. It is not clear whether treatment directed at high triglyceride levels will reduce risk for initial or recurrent CHD events. Also, triglyceride measurements vary considerably for individual patients. Accordingly, the ATP III (2) provides guidance for the management of elevated triglyceride levels by focusing on a combination of therapeutic lifestyle changes and by a secondary lipid target for non-HDL cholesterol.

#### E. Obesity

Obesity is a common condition associated with increased risk for coronary disease and mortality (18). Obesity is defined as a body mass index (weight in kilograms divided by the square of height in meters) of 30 kg/m², and overweight begins at 25 kg/m² (19). Obesity is associated with and contributes to other coronary disease risk factors, including high blood pressure, glucose intolerance, low HDL cholesterol, and elevated triglyceride levels. Hence, much of the increased CAD risk associated with obesity is mediated by these risk factors. Risk is particularly raised in the presence of abdominal obesity, which can be identified by a waist circumference greater than 102 cm (40 inches) in men or 88 cm (35 inches) in women (19). Because weight reduction in overweight and obese people is a method to reduce multiple other risk factors, it is an important component of secondary prevention of CHD.

#### F. Metabolic Syndrome

Evidence is accumulating that risk for future CHD events can be reduced beyond that achieved by LDL-lowering therapy by modification of a specific secondary target of therapy—the metabolic syndrome—represented by a constellation of lipid and nonlipid risk factors of metabolic origin. This syndrome is closely linked to a generalized metabolic disorder called insulin resistance in which the normal actions of insulin are impaired. Excess body fat (particularly abdominal obesity) and physical inactivity promote the development of insulin resistance, but some individuals also are genetically predisposed to insulin resistance. In a field that has been confused by nomenclature, ATP III has introduced a standard definition for the diagnosis of the metabolic syndrome, as shown in the Table. The metabolic syndrome is considered to be present when 3 or more of the characteristics are present.

Management of the metabolic syndrome has a 2-fold objective: (1) to reduce underlying causes (ie, obesity and physical inactivity) and (2) to treat associated nonlipid and lipid risk factors. First-line therapies for all lipid and nonlipid risk factors associated with the metabolic syndrome are weight reduction and increased physical activity, as well as appropriate control of LDL cholesterol. In patients with triglycerides greater than 200 mg/dL, a non-HDL cholesterol goal of less than 130 mg/dL is a secondary target (see the Table).

#### G. Hormone Replacement Therapy

On the basis of epidemiological studies and prospective studies of the effects of estrogen administration on cardiac

#### **Characteristics Used to Define Metabolic Syndrome**

Risk Factor	Defining Level
Abdominal obesity	Waist circumference
Men	Greater than 103 cm (40 in)
Women	Greater than 88 cm (35 in)
Triglycerides	Greater than 150 mg/dL
HDL cholesterol	
Men	Less than 40 mg/dL
Women	Less than 50 mg/dL
Blood pressure	Greater than or equal to 130/85 mm Hg
Fasting serum glucose	Greater than or equal to 110 mg/dL

risk factors, postmenopausal estrogen replacement has been advocated for both primary and secondary prevention of CAD in women. However, the first published randomized trial of estrogen plus progestin therapy in postmenopausal women with known CAD did not show any reduction in cardiovascular events over 4 years of follow-up (20), despite an 11% lower LDL cholesterol level and a 10% higher HDL cholesterol level in those women receiving HRT. In addition, women receiving HRT had higher rates of cardiovascular events during the first 2 years, more thromboembolic events, and more gallbladder disease (20). A subsequent angiographic study also revealed no benefit from HRT (21). Another prospective randomized controlled trial with HRT in women with a history of stroke found no benefit in reducing mortality or stroke after 2.8 years (22). The Women's Health Initiative, a randomized controlled primary prevention trial of estrogen plus progestin, found that the overall health risks of this therapy exceeded its benefits (23). Thus, current information suggests that HRT in postmenopausal women does not reduce risk for major vascular events or coronary deaths in secondary prevention. Women who are taking HRT and who have vascular disease can continue this therapy if it is being prescribed for other well-established indications and no better alternative therapies are appropriate. However, at the present time, there is no basis for adding or continuing estrogens in postmenopausal women with clinically evident CAD or cerebrovascular disease in an effort to prevent or retard progression of their underlying disease (24).

If a woman develops an acute CAD event while undergoing HRT, it is prudent to consider discontinuance of the HRT (24). In women who are immobilized, HRT should be discontinued, or venous thromboembolism prophylaxis should be used (24).

#### H. Oxidative Stress

Evidence from clinical trials is negative regarding the effects of supplementation with antioxidant vitamins. Although several small trials and in vitro data from basic research in vascular biology have suggested that vitamin C and/or E might interfere with formation of atherosclerotic lesions, 2 large randomized clinical trials have shown no benefit when vitamin E was given to post-MI patients (25) or in those with vascular disease or diabetics with a high-risk CAD profile (26–28). Furthermore, a small coronary regression trial, the HDL Atherosclerosis Treatment Study (HATS), suggested an

adverse effect of antioxidant vitamins on coronary atherosclerosis, clinical events, and HDL and apoA-1 metabolism (27,28). The use of vitamin E (administered with vitamin C and beta-carotene) was the subject of a large (more than 20 000 participants) trial of patients at risk for CAD and with CAD (26). Antioxidant therapy had no effect on the end points of cardiovascular death, cardiovascular events, stroke, or revascularization, considered alone or in combination. Although previous observational and epidemiological studies have suggested a benefit from dietary supplementation with antioxidants or a diet rich in antioxidants, especially vitamin E, there is currently no basis for recommending that patients take vitamin C or E supplements or other antioxidants for the express purpose of preventing or treating CAD.

#### IV. Alternative Therapies for Chronic Stable Angina in Patients Refractory to Medical Therapy Who Are Not Candidates for Percutaneous Intervention or Revascularization

#### Recommendations

Class IIa

1. Surgical laser transmyocardial revascularization (TMR). (Level of Evidence: A)

Class IIb

- 1. Enhanced external counterpulsation (EECP). (Level of Evidence: B)
- 2. Spinal cord stimulation (SCS). (Level of Evidence: B)

Evidence has emerged regarding the relative efficacy, or lack thereof, of a number of techniques for the management of refractory chronic angina pectoris. These techniques should only be used in patients who cannot be managed adequately by medical therapy and who are not candidates for revascularization (interventional and/or surgical).

#### A. Spinal Cord Stimulation

Since approximately 1987, SCS has been proposed as a method for providing analgesia for patients with chronic angina pectoris refractory to medical, catheter interventional, or surgical therapy. The efficacy of SCS depends on the accurate placement of the stimulating electrode in the dorsal epidural space, usually at the C7-T1 level. A review of the literature has revealed 2 small randomized clinical trials involving implanted spinal cord stimulators, 1 of which directly tested its efficacy. One report studied the efficacy of SCS in 13 treated patients versus 12 control subjects, both groups with chronic intractable angina pectoris studied for 6 weeks (29). Another small randomized trial involved 24 patients with refractory angina (30). Nine other studies, either retrospective (31–33) or prospective (29,34–38) cohort studies, were identified in the literature. Although the results of these studies appear promising, there is still a paucity of data on the intermediate- and long-term benefit of these devices.

#### B. Enhanced External Counterpulsation

Another nonpharmacological technique that has been described for treatment of patients with chronic stable angina is known as EECP. EECP was evaluated in a randomized, placebo-controlled multicenter trial to determine its safety and efficacy (39). Patients (n=139) with chronic stable angina, documented CAD, and a positive exercise treadmill test were randomly assigned to receive EECP (35 hours of active counterpulsation) or inactive EECP over a 4- to 7-week period. The authors concluded that EECP decreased angina frequency (P < 0.05) and improved time to exercise-induced ischemia (P=0.01). Two multicenter registry studies that included 978 patients from 43 centers (40) and 2289 patients from more than 100 centers (41) evaluated the safety and effectiveness of EECP in treating chronic stable angina. These studies found the treatment to be generally well tolerated and efficacious; anginal symptoms were improved in approximately 75% to 80% of patients. However, additional clinical trial data are necessary before this technology can be recommended definitively.

#### C. Laser TMR

Another emerging technique that has been studied for the treatment of more severe chronic stable angina refractory to medical or other therapies is laser TMR. This technique has either been performed in the operating room (with a carbon dioxide or holmium:YAG laser) or by a percutaneous approach with a specialized (holmium:YAG laser) catheter. Eight prospective randomized clinical trials have been performed, 2 with the percutaneous technique and the other 6 with an epicardial surgical technique (42–48). The goal in both approaches is to create a series of transmural endomyocardial channels to improve myocardial revascularization. Percutaneous TMR technology has not been approved by the Food and Drug Administration and should therefore be considered an experimental therapy.

The surgical TMR technique has generally been associated with improvement in symptoms in patients with chronic stable angina. The mechanism for improvement in anginal symptoms is still controversial. Possible mechanisms for this improvement include increased myocardial perfusion, denervation of the myocardium, stimulation of angiogenesis, or perhaps some other unknown mechanism (49–51). There are conflicting data regarding improvement in exercise capacity (43,44,47). Despite the apparent benefit in decreasing anginal symptoms, no definite benefit has been demonstrated in terms of increasing myocardial perfusion (43,46,47).

There are currently no published studies to document the long-term efficacy of surgical TMR. Nonetheless, this technique appears to provide symptomatic relief for end-stage chronic angina in the short term. Additional follow-up studies are necessary to evaluate procedural efficacy in patients who have undergone surgical laser TMR alone, as well as coronary bypass surgery plus TMR.

## V. Asymptomatic Patients With Known or Suspected CAD

At the direction of the Task Force, this update outlines the approach to asymptomatic patients with known or suspected

CAD on the basis of a history and/or electrocardiographic (ECG) evidence of previous MI, coronary angiography, or an abnormal noninvasive test. The inclusion of asymptomatic patients with abnormal noninvasive tests does not constitute an endorsement of such tests for the purposes of screening but simply acknowledges the clinical reality that such patients often present for evaluation after such tests have been performed. Multiple ACC/AHA guidelines and scientific statements have discouraged the use of ambulatory monitoring, treadmill testing, stress echocardiography, stress myocardial perfusion imaging, and electron-beam computed tomography (EBCT) as routine screening tests in asymptomatic individuals.

The full-text guideline includes many subsections on asymptomatic patients, which appear at the end of the appropriate sections for symptomatic patients. They are integrated here into a single summary. The recommendations for noninvasive testing, which follow, appear in multiple different lists of recommendations in the full-length document.

#### Recommendations for Noninvasive Testing for the Diagnosis of Obstructive CAD and Risk Stratification in Asymptomatic Patients Class IIb

- 1. Exercise ECG testing without an imaging modality in asymptomatic patients with possible myocardial ischemia on ambulatory ECG (AECG) monitoring or with severe coronary calcification on EBCT in the absence of one of the following ECG abnormalities:
  - a. Preexcitation (Wolff-Parkinson-White) syndrome (Level of Evidence: C)
  - b. Electronically paced ventricular rhythm (Level of Evidence: C)
  - c. More than 1 mm of ST depression at rest (Level of Evidence: C)
  - d. Complete left bundle-branch block. (Level of Evidence: C)
- 2. Exercise perfusion imaging or exercise echocardiography in asymptomatic patients with possible myocardial ischemia on AECG monitoring or with severe coronary calcification on EBCT who are able to exercise and have one of the following baseline ECG abnormalities:
  - a. Preexcitation (Wolff-Parkinson-White) syndrome (Level of Evidence: C)
  - b. More than 1 mm of ST depression at rest. (Level of Evidence: C)
- 3. Adenosine or dipyridamole myocardial perfusion imaging in patients with severe coronary calcification on EBCT but with one of the following baseline ECG abnormalities:
  - a. Electronically paced ventricular rhythm (Level of Evidence: C)
  - b. Left bundle-branch block. (Level of Evidence: C)
- 4. Adenosine or dipyridamole myocardial perfusion imaging or dobutamine echocardiography in patients with possible myocardial ischemia on AECG

- monitoring or with coronary calcification on EBCT who are unable to exercise. (Level of Evidence: C)
- 5. Exercise myocardial perfusion imaging or exercise echocardiography after exercise ECG testing in asymptomatic patients with an intermediate-risk or high-risk Duke treadmill score. (Level of Evidence: C)
- 6. Adenosine or dipyridamole myocardial perfusion imaging or dobutamine echocardiography after exercise ECG testing in asymptomatic patients with an inadequate exercise ECG. (Level of Evidence: C)

#### Class III

- 1. Exercise ECG testing without an imaging modality in asymptomatic patients with possible myocardial ischemia on AECG monitoring or with coronary calcification on EBCT but with the baseline ECG abnormalities listed under Class IIb¹ above. (Level of Evidence: B)
- 2. Exercise ECG testing without an imaging modality in asymptomatic patients with an established diagnosis of CAD owing to prior MI or coronary angiography; however, testing can assess functional capacity and prognosis. (Level of Evidence: B)
- 3. Exercise or dobutamine echocardiography in asymptomatic patients with left bundle-branch block. (Level of Evidence: C)
- 4. Adenosine or dipyridamole myocardial perfusion imaging or dobutamine echocardiography in asymptomatic patients who are able to exercise and who do not have left bundle-branch block or electronically paced ventricular rhythm. (Level of Evidence: C)
- 5. Exercise myocardial perfusion imaging, exercise echocardiography, adenosine or dipyridamole myocardial perfusion imaging, or dobutamine echocardiography after exercise ECG testing in asymptomatic patients with a low-risk Duke treadmill score. (Level of Evidence: C)

The use of exercise ECG testing in asymptomatic patients as a means of screening for CAD is discussed in detail in the ACC/AHA Guideline Update for Exercise Testing (52).

In the absence of symptoms, AECG monitoring can reveal transient ST-segment depression suggestive of CAD. However, as indicated in the ACC/AHA Guidelines for Ambulatory Electrocardiography (53), there is presently no evidence that AECG monitoring provides reliable information concerning ischemia in asymptomatic subjects without known CAD.

In the absence of symptoms, EBCT is sometimes used as a means of screening for CAD. However, as indicated in the ACC/AHA Expert Consensus Document on Electron-Beam Computed Tomography for the Diagnosis and Prognosis of Coronary Artery Disease (54), available data are insufficient to support recommending EBCT for this purpose to asymptomatic members of the general public.

Physicians are often confronted with concerned asymptomatic patients with abnormal findings on AECG and EBCT. Although the published data on this situation are scant, in the absence of symptoms, such patients probably still have a low pretest probability of significant CAD. A negative exercise test result only confirms the low probability of disease, and a positive test result may not increase the probability of disease enough to make a clinical difference.

Stress imaging procedures (ie, either stress myocardial perfusion imaging or stress echocardiography) are generally not indicated as the initial stress test in most such patients. However, in patients with resting ECG abnormalities that preclude adequate interpretation of the exercise ECG, stress imaging procedures are preferable to the exercise ECG. If the baseline ECG shows preexcitation or more than 1 mm of ST depression, exercise stress imaging procedures are preferred. If the resting ECG shows a ventricularly paced rhythm or left bundle-branch block, vasodilator perfusion imaging is preferred. In patients who are unable to exercise, pharmacological stress imaging is preferable to exercise ECG testing. The preference for stress imaging under these circumstances is based on the available literature in symptomatic patients. There are scant published data on the use of stress imaging procedures in asymptomatic patients in general and in particular on asymptomatic patients with resting ECG abnormalities or asymptomatic patients who are unable to exercise. In asymptomatic patients with an intermediate-risk or high-risk Duke treadmill score on exercise ECG testing, stress imaging procedures are potentially useful as a second diagnostic test. Given the low pretest probability of asymptomatic patients, an abnormal exercise ECG in such a patient is likely a false-positive that will be confirmed by a negative stress image. However, the published data demonstrating the efficacy of stress imaging procedures in these specific circumstances are scant. In the presence of a low-risk Duke treadmill score on exercise ECG testing, stress imaging procedures in asymptomatic patients are usually not justified.

In asymptomatic patients, risk stratification and prognosis are more important considerations than diagnosis. Because the treatment of asymptomatic patients cannot improve their symptoms, the principal goal of evaluation and treatment is the improvement of patient outcome by reducing the rate of death and nonfatal MI. In one large study dominated by asymptomatic patients (55), the Duke treadmill score predicted subsequent cardiac events. However, the absolute event rate was low, even in patients with high-risk scores, which suggests that the ability to improve outcome with revascularization in such patients is limited.

The prognostic value of stress imaging procedures in asymptomatic patients is not well established. Some of the published series did include asymptomatic patients. However, this subset of patients was generally not analyzed separately. Blumenthal et al (56) reported a small study using exercise thallium testing in siblings of patients with premature coronary atherosclerosis. They demonstrated that the combination of an abnormal exercise ECG and a positive thallium image was prognostically important. However, many of the events included in their analysis were subsequent revascularizations, the performance of which was clearly influenced by the results of the exercise thallium test. Given the generally low event rate in asymptomatic patients, the ability of stress imaging procedures to identify a subset with a substantial absolute risk of subsequent events is problematic, with the possible exception of patients with previous MI.

#### Recommendations for Coronary Angiography for Risk Stratification in Asymptomatic Patients Class IIa

1. Patients with high-risk criteria that suggest ischemia on noninvasive testing. (Level of Evidence: C)

#### Class IIb

1. Patients with inadequate prognostic information after noninvasive testing. (Level of Evidence: C)

#### Class III

1. Patients who prefer to avoid revascularization. (Level of Evidence: C)

The noninvasive test findings that identify high-risk patients are based on studies in symptomatic patients. These findings are probably also applicable to asymptomatic patients but associated with a lower level of absolute risk in the absence of symptoms. The mere presence of left ventricular dysfunction in an asymptomatic patient probably does not justify coronary angiography. However, other high-risk noninvasive test findings that reflect myocardial ischemia, such as a high-risk Duke treadmill score, a large stress-induced perfusion defect, or an extensive echocardiographic wall-motion abnormality that develops at a low heart rate, are probably appropriate indications for coronary angiography, although there are only limited data to support this approach. The ability to improve outcome in such patients has not been demonstrated.

#### Recommendations for Pharmacotherapy to Prevent MI and Death in Asymptomatic Patients Class I

- 1. Aspirin in the absence of contraindication in patients with prior MI. (Level of Evidence: A)
- 2. Beta-blockers as initial therapy in the absence of contraindications in patients with prior MI. (Level of Evidence: B)
- 3. Lipid-lowering therapy in patients with documented CAD and LDL cholesterol greater than 130 mg/dL, with a target LDL of less than 100 mg/dL. (Level of Evidence: A)
- 4. ACE inhibitor in patients with CAD\* who also have diabetes and/or left ventricular systolic dysfunction. (Level of Evidence: A)

#### Class IIa

- 1. Aspirin in the absence of contraindications in patients without prior MI. (Level of Evidence: B)
- 2. Beta-blockers as initial therapy in the absence of contraindications in patients without prior MI. (Level of Evidence: C)
- 3. Lipid-lowering therapy in patients with documented CAD and LDL cholesterol 100 to 129 mg/dL, with a target LDL of 100 mg/dL. (Level of Evidence: C)
- 4. ACE inhibitor in all patients with CAD\* or other vascular disease. (Level of Evidence: B)

#### \*Significant CAD by angiography or previous MI.

Even in asymptomatic patients, aspirin and beta-blockers are recommended in patients with prior MI. The data in support of these recommendations are detailed in the ACC/AHA

Guidelines for the Management of Patients With Acute Myocardial Infarction (57). In the absence of prior MI, patients with documented CAD on the basis of noninvasive testing or coronary angiography probably also benefit from aspirin, although the data on this specific subset of patients are limited.

Several studies have investigated the potential role of  $\beta$ -blockers in patients with asymptomatic ischemia demonstrated on exercise testing and/or ambulatory monitoring. The data generally demonstrate a benefit from  $\beta$ -blocker therapy, but not all trials have been positive.

Lipid-lowering therapy in asymptomatic patients with documented CAD was demonstrated to decrease the rate of adverse ischemic events in 4S (58), as well as in the CARE study (59), as mentioned previously.

#### A. Treatment of Risk Factors

In asymptomatic patients with documented CAD on the basis of noninvasive testing or coronary angiography, the treatment of risk factors outlined above is clearly appropriate. The recommendations for symptomatic patients should be followed as detailed in the 1999 guideline and the update section above.

In the absence of documented CAD, asymptomatic patients should also undergo treatment of risk factors according to primary prevention standards. Therapy should be directed toward hypertension, smoking cessation, diabetes, exercise training, and weight reduction in the presence of other risk factors. Lipid-lowering therapy should be administered according to the primary prevention standards outlined in the 2001 National Cholesterol Education Program guidelines.<sup>2</sup>

#### B. Revascularization

In asymptomatic patients, revascularization cannot improve symptoms. The only appropriate indication for revascularization with either percutaneous coronary intervention or coronary artery bypass grafting is therefore to improve prognosis. Most of the recommendations for revascularization that were published in 1999 for patients with stable angina also apply to asymptomatic patients, because their underlying rationale is to improve prognosis. The single Class I recommendation for revascularization in patients who have not been successfully treated by medical therapy is an exception and obviously does not apply to asymptomatic patients. However, the level of evidence in support of these recommendations in asymptomatic patients is clearly weaker than in symptomatic patients. Most of the available randomized trial data have focused on symptomatic patients. Their extrapolation to asymptomatic patients appears reasonable but is based on far more limited evidence.

In the Coronary Artery Surgery Study Registry (60), asymptomatic patients with left main CAD who underwent coronary artery bypass grafting had a better outcome than those patients treated with medical therapy, but this was not a randomized trial. The most compelling randomized trial data on asymptomatic patients comes from the previously mentioned Asymptomatic Cardiac Ischemia Pilot (ACIP) study (61,62). In patients with CAD who were either free of angina or had well-controlled symptoms, patients randomized to revascularization had a lower cardiac event rate than patients who were randomized to medical management

guided by angina or medical management guided by noninvasive ischemia. The patients entered in this study, who were required to have ischemia during ambulatory monitoring and exercise testing, as well as significant CAD, were more likely to have extensive CAD and prior MI. In the overall study group, 39% of the patients had 3-vessel disease, 40% had prior MI, 22% had prior revascularization, and 59% had angina within the previous 6 weeks. Many of the patients enrolled in this trial presumably came to medical attention because of symptoms or prior MI. The degree to which the results of ACIP can be applied to patients who have never been symptomatic and have less severe asymptomatic CAD is uncertain.

#### References

- 1. Yusuf S, Sleight P, Pogue J, et al. Effects of an angiotensinconverting-enzyme inhibitor, ramipril, on cardiovascular events in high-risk patients. The Heart Outcomes Prevention Evaluation Study Investigators. N Engl J Med 2000;342:145–53.
- 2. Executive Summary of the Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). JAMA 2001;285:2486-97.
- 2a.MRC/BHF Heart Protection Study of cholesterol lowering with simvastatin in 20,536 high-risk individuals: a randomised placebocontrolled trial. Lancet 2002;360:7-22.
- Vega GL, Grundy SM. Primary hypertriglyceridemia with borderline high cholesterol and elevated apolipoprotein B concentrations: comparison of gemfibrozil vs lovastatin therapy. JAMA 1990;264:2759–63.
- Abate N, Vega GL, Grundy SM. Variability in cholesterol content and physical properties of lipoproteins containing apolipoprotein B-100. Atherosclerosis 1993;104:159–71.
- Sniderman AD. Apolipoprotein B and apolipoprotein AI as predictors of coronary artery disease. Can J Cardiol 1988;4 Suppl A:24A–30A.
- Reinhart RA, Gani K, Arndt MR, et al. Apolipoproteins A-I and B as predictors of angiographically defined coronary artery disease. Arch Intern Med 1990;150:1629–33.
- Sniderman A, Vu H, Cianflone K. Effect of moderate hypertriglyceridemia on the relation of plasma total and LDL apo B levels. Atherosclerosis 1991;89:109–16.
- Tornvall P, Bavenholm P, Landou C, et al. Relation of plasma levels and composition of apolipoprotein B-containing lipoproteins to angiographically defined coronary artery disease in young patients with myocardial infarction. Circulation 1993;88:2180–9.
- Westerveld HT, van Lennep JE, van Lennep HW, et al. Apolipoprotein B and coronary artery disease in women: a cross-sectional study in women undergoing their first coronary angiography. Arterioscler Thromb Vasc Biol 1998;18:1101–7.
- Lamarche B, Moorjani S, Lupien PJ, et al. Apolipoprotein A-I and B levels and the risk of ischemic heart disease during a five-year follow-up of men in the Quebec cardiovascular study. Circulation 1996;94:273–8.
- Gotto AM Jr, Whitney E, Stein EA, et al. Relation between baseline and on-treatment lipid parameters and first acute major coronary events in the Air Force/Texas Coronary Atherosclerosis Prevention Study (AFCAPS/ TexCAPS). Circulation 2000;101:477–84.
- 12. Lemieux I, Pascot A, Couillard C, et al. Hypertriglyceridemic waist: a marker of the atherogenic metabolic triad (hyperinsulinemia; hyperapolipoprotein B; small, dense LDL) in men? Circulation 2000; 102:179-84.
- 13. Jeppesen J, Hein HO, Suadicani P, et al. Triglyceride concentration and ischemic heart disease: an eight-year follow-up in the Copenhagen Male Study. Circulation 1998;97:1029–36.
- Reaven GM. Insulin resistance and compensatory hyperinsulinemia: role in hypertension, dyslipidemia, and coronary heart disease. Am Heart J 1991;121:1283–8.
- Grundy SM, Vega GL. Two different views of the relationship of hypertriglyceridemia to coronary heart disease: implications for treatment. Arch Intern Med 1992;152:28–34.
- Assmann G, Schulte H, Funke H, et al. The emergence of triglycerides as a significant independent risk factor in coronary artery disease. Eur Heart J 1998;19 Suppl M:M8-14.

- Austin MA, Hokanson JE, Edwards KL. Hypertriglyceridemia as a cardiovascular risk factor. Am J Cardiol 1998;81:7B–12B.
- Hubert HB, Feinleib M, McNamara PM, et al. Obesity as an independent risk factor for cardiovascular disease: a 26-year follow-up of participants in the Framingham Heart Study. Circulation 1983;67:968–77.
- Clinical guidelines on the identification, evaluation, and treatment of overweight and obesity in adults—the evidence report: National Institutes of Health. Obes Res 1998;6 Suppl 2:51S–209S.
- Hulley S, Grady D, Bush T, et al, for the Heart and Estrogen/progestin Replacement Study (HERS) Research Group. Randomized trial of estrogen plus progestin for secondary prevention of coronary heart disease in postmenopausal women. JAMA 1998;280:605–13.
- Herrington DM, Reboussin DM, Brosnihan KB, et al. Effects of estrogen replacement on the progression of coronary-artery atherosclerosis. N Engl J Med 2000;343:522-9.
- Viscoli CM, Brass LM, Kernan WN, et al. A clinical trial of estrogenreplacement therapy after ischemic stroke. N Engl J Med 2001;345: 1243–9.
- Risks and benefits of estrogen plus progestin in healthy postmenopausal women: principal results from the Women's Health Initiative randomized controlled trial. JAMA 2002;288:321-33.
- Mosca L, Collins P, Herrington DM, et al. Hormone replacement therapy and cardiovascular disease: a statement for healthcare professionals from the American Heart Association. Circulation 2001; 104:499-503.
- 25. Gruppo Italiano per lo Studio della Sopravvivenza nell'Infarto miocardico. Dietary supplementation with n-3 polyunsaturated fatty acids and vitamin E after myocardial infarction: results of the GISSI-Prevenzione trial. Lancet 1999;354:447-55.
- Heart Protection Study Collaborative Group. MRC/BHF Heart Protection Study of antioxidant vitamin supplementation in 20 536 high-risk individuals: a randomised placebo-controlled trial. Lancet 2002;360:22–33.
- 27. Cheung MC, Zhao XQ, Chait A, et al. Antioxidant supplements block the response of HDL to simvastatin-niacin therapy in patients with coronary artery disease and low HDL. Arterioscler Thromb Vasc Biol 2001;21:1320-6.
- Brown BG, Zhao XQ, Chait A, et al. Simvastatin and niacin, antioxidant vitamins, or the combination for the prevention of coronary disease. N Engl J Med 2001;345:1583–92.
- Hautvast RW, DeJongste MJ, Staal MJ, et al. Spinal cord stimulation in chronic intractable angina pectoris: a randomized, controlled efficacy study. Am Heart J 1998;136:1114–20.
- 30. Jessurun GA, DeJongste MJ, Hautvast RW, et al. Clinical follow-up after cessation of chronic electrical neuromodulation in patients with severe coronary artery disease: a prospective randomized controlled study on putative involvement of sympathetic activity. Pacing Clin Electrophysiol 1999;22:1432–9.
- Jessurun GA, Ten Vaarwerk IA, DeJongste MJ, et al. Sequelae of spinal cord stimulation for refractory angina pectoris: reliability and safety profile of long-term clinical application. Coron Artery Dis 1997;8:33–8.
- 32. TenVaarwerk IA, Jessurun GA, DeJongste MJ, et al, for the Working Group on Neurocardiology. Clinical outcome of patients treated with spinal cord stimulation for therapeutically refractory angina pectoris. Heart 1999;82:82-8.
- Murray S, Carson KG, Ewings PD, et al. Spinal cord stimulation significantly decreases the need for acute hospital admission for chest pain in patients with refractory angina pectoris. Heart 1999;82: 89-92.
- De Landsheere C, Mannheimer C, Habets A, et al. Effect of spinal cord stimulation on regional myocardial perfusion assessed by positron emission tomography. Am J Cardiol 1992;69:1143–9.
- Eliasson T, Albertsson P, Hardhammar P, et al. Spinal cord stimulation in angina pectoris with normal coronary arteriograms. Coron Artery Dis 1993;4:819–27.
- de Jongste MJ, Nagelkerke D, Hooyschuur CM, et al. Stimulation characteristics, complications, and efficacy of spinal cord stimulation systems in patients with refractory angina: a prospective feasibility study. Pacing Clin Electrophysiol 1994;17:1751–60.
- Hautvast RW, Blanksma PK, DeJongste MJ, et al. Effect of spinal cord stimulation on myocardial blood flow assessed by positron emission tomography in patients with refractory angina pectoris. Am J Cardiol 1996;77:462–7.

- 38. Greco S, Auriti A, Fiume D, et al. Spinal cord stimulation for the treatment of refractory angina pectoris: a two-year follow-up. Pacing Clin Electrophysiol 1999;22:26–32.
- Arora RR, Chou TM, Jain D, et al. The multicenter study of enhanced external counterpulsation (MUST-EECP): effect of EECP on exercise-induced myocardial ischemia and anginal episodes. J Am Coll Cardiol 1999;33:1833–40.
- Barsness G, Feldman AM, Holmes DR Jr, et al. The International EECP Patient Registry (IEPR): design, methods, baseline characteristics, and acute results. Clin Cardiol 2001;24:435–42.
- Lawson WE, Hui JC, Lang G. Treatment benefit in the enhanced external counterpulsation consortium. Cardiology 2000:94:31–5.
- 42. Oesterle SN, Sanborn TA, Ali N, et al. Percutaneous transmyocardial laser revascularisation for severe angina: the PACIFIC randomised trial: Potential Class Improvement From Intramyocardial Channels. Lancet 2000;356:1705–10.
- 43. Burkhoff D, Schmidt S, Schulman SP, et al. Transmyocardial laser revascularisation compared with continued medical therapy for treatment of refractory angina pectoris: a prospective randomised trial: ATLANTIC Investigators: Angina Treatments-Lasers and Normal Therapies in Comparison. Lancet 1999;354:885–90.
- 44. Aaberge L, Nordstrand K, Dragsund M, et al. Transmyocardial revascularization with CO2 laser in patients with refractory angina pectoris: clinical results from the Norwegian randomized trial. J Am Coll Cardiol 2000;35:1170-7.
- 45. Held C, Hjemdahl P, Hakan WN, et al. Inflammatory and hemostatic markers in relation to cardiovascular prognosis in patients with stable angina pectoris: results from the APSIS study: the Angina Prognosis Study in Stockholm. Atherosclerosis 2000;148:179–88.
- Allen KB, Dowling RD, Fudge TL, et al. Comparison of transmyocardial revascularization with medical therapy in patients with refractory angina. N Engl J Med 1999;341:1029-36.
- 47. Schofield PM, Sharples LD, Caine N, et al. Transmyocardial laser revascularisation in patients with refractory angina: a randomised controlled trial. Lancet 1999;353:519-24.
- Frazier OH, March RJ, Horvath KA. Transmyocardial revascularization with a carbon dioxide laser in patients with end-stage coronary artery disease. N Engl J Med 1999;341:1021–8.
- Horvath KA, Smith WJ, Laurence RG, et al. Recovery and viability of an acute myocardial infarct after transmyocardial laser revascularization. J Am Coll Cardiol 1995;25:258-63.
- Yamamoto N, Kohmoto T, Gu A, et al. Angiogenesis is enhanced in ischemic canine myocardium by transmyocardial laser revascularization. J Am Coll Cardiol 1998;31:1426–33.
- Kwong KF, Kanellopoulos GK, Nickols JC, et al. Transmyocardial laser treatment denervates canine myocardium. J Thorac Cardiovasc Surg 1997:114:883-9.
- 52. Gibbons RJ, Balady GJ, Bricker JT, et al. ACC/AHA 2002 guideline update for exercise testing: a report of the American College of Cardiology/American Heart Association Task Force on Practice

- Guidelines (Committee on Exercise Testing). 2002. American College of Cardiology Web site. Available at: http://www.acc.org/clinical/guidelines/exercise/exercise\_clean.pdf. Accessed October 17, 2002.
- 53. Crawford MH, Bernstein SJ, Deedwania PC, et al. ACC/AHA guidelines for ambulatory electrocardiography: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee to Revise the Guidelines for Ambulatory Electrocardiography): developed in collaboration with the North American Society for Pacing and Electrophysiology. J Am Coll Cardiol 1999;34:912–48.
- 54. O'Rourke RA, Brundage BH, Froelicher VF, et al. American College of Cardiology/American Heart Association expert consensus document on electron-beam computed tomography for the diagnosis and prognosis of coronary artery disease. J Am Coll Cardiol 2000; 36:326-40.
- Nishime EO, Cole CR, Blackstone EH, et al. Heart rate recovery and treadmill exercise score as predictors of mortality in patients referred for exercise ECG. JAMA 2000;284:1392–8.
- Blumenthal JA, Jiang W, Babyak MA, et al. Stress management and exercise training in cardiac patients with myocardial ischemia: effects on prognosis and evaluation of mechanisms. Arch Intern Med 1997;157: 2213–23.
- 57. Ryan TJ, Antman EM, Brooks NH, et al. ACC/AHA guidelines for the management of patients with acute myocardial infarction: 1999 update: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (Committee on Management of Acute Myocardial Infarction). 1999. American College of Cardiology Web site. Available at: http://www.acc.org/ clinical/guidelines/nov96/1999/index.htm. Accessed October 17, 2002.
- Randomised trial of cholesterol lowering in 4444 patients with coronary heart disease: the Scandinavian Simvastatin Survival Study (4S). Lancet 1994;344:1383–9.
- Sacks FM, Pfeffer MA, Moye LA, et al, for the Cholesterol and Recurrent Events Trial Investigators. The effect of pravastatin on coronary events after myocardial infarction in patients with average cholesterol levels. N Engl J Med 1996;335:1001–9.
- Taylor HA, Deumite NJ, Chaitman BR, et al. Asymptomatic left main coronary artery disease in the Coronary Artery Surgery Study (CASS) registry. Circulation 1989;79:1171–9.
- Davies RF, Goldberg AD, Forman S, et al. Asymptomatic Cardiac Ischemia Pilot (ACIP) study two-year follow-up: outcomes of patients randomized to initial strategies of medical therapy versus revascularization. Circulation 1997;95:2037–43.
- 62. Sharaf BL, Williams DO, Miele NJ, et al. A detailed angiographic analysis of patients with ambulatory electrocardiographic ischemia: results from the Asymptomatic Cardiac Ischemia Pilot (ACIP) study angiographic core laboratory. J Am Coll Cardiol 1997;29:78–84.

KEY WORDS: ACC/AHA Guideline Update ■ angina ■ risk factors ■ coronary disease ■ revascularization